Author’s response to reviews

Title: Guided self-help on the Internet for Turkish migrants with depression: the design of a randomized controlled trial

Authors:

Burçin Ünlü (b.unlu@psy.vu.nl)

Version: 5 Date: 19 October 2010

Author’s response to reviews: see over
Amsterdam, 2010, October 19

Dear Editor,

We would like to submit the revised version of the following manuscript for publication in the journal of Trials: Guided self-help on the Internet for Turkish migrants with depression: the design of a randomized controlled trial, by Burçin Ünlü, Heleen Riper, Annemieke van Straten and Pim Cuijpers.

The revisions include a legend section and a figure title after the reference list. The Trial Registration number has also been changed in NTR2303.

We look forward to hearing whether you can accept this manuscript for publication.

Yours sincerely,

Burçin Ünlü
Department of Clinical Psychology
Van der Boechorststraat 1, 1081 BT Amsterdam
VU University, the Netherlands
Tel: (+31) 020 – 598 36 88
Fax: (+31) 020 – 598 87 58
E-Mail: b.unlu@psy.vu.nl